Author's response to reviews

Title: Voices from the frontline: Counsellors' perspectives on TB/HIV collaborative activities in the Northwest Region, Cameroon.

Authors:

Barnabas Nwarbébé Njozing (njozing.barnabas@epiph.umu.se)
Kerstin E Edin (kerstin.edin@epiph.umu.se)
San Sebastián Miguel (miguel.sansebastian@epiph.umu.se)
Anna-Karin Hurtig (anna-karin.hurtig@epiph.umu.se)

Version: 4 Date: 22 November 2011

Author's response to reviews: see over
POINT-BY-POINT DESCRIPTION OF FORMATTTING CHANGES

1. Author list
   The list of authors in the manuscript has been re-written as they appear in the submission system both in style and order.

2. Competing interest
   The subsection on competing interest has been reworded to ‘The authors declare that they have no competing interests’.

3. Highlighting/tracking
   All highlighting and tracking have been removed from the manuscript.

4. Page numbers
   Page numbers have been removed from the manuscript.

5. Typography
   The entire manuscript has been read properly; typographical errors and relevant revisions have been effected.